Author’s response to reviews

Title: ADEQUACY OF CONTROL OF CARDIOVASCULAR RISK FACTORS IN AMBULATORY PATIENTS WITH TYPE 2 DIABETES ATTENDING DIABETES OUTPATIENT CLINIC AT A COUNTY HOSPITAL, KENYA

Authors:

Frederick C F Otieno (cfotieno@gmail.com)
Elijah Ogola (onogola@gmail.com)
Mercy Kimando (mkimando@gmail.com)
Kenn Mutai (mutaikk@gmail.com)

Version: 1 Date: 12 Jun 2017

Author’s response to reviews:

Response to the Reviewers are incorporated in the text of the revised manuscript but written in BOLD.

Context of the study has been explained under methods.

More references from the scarce but relevant studies, reviews/publications from sub-Saharan Africa have been added. The text is a bit longer but the various cardiovascular risk factors in our study and the activities around controlling them, jointly and severally demanded referencing.

Conclusion has been revised.